Author’s response to reviews

Title: The Chinese version of Attitudes towards guidelines scale: Validity and reliability assessment

Authors:

Wen Yan (381590064@qq.com)

Qian Chen (hxqianchen@126.com)

Xue Zhang (2041650433@qq.com)

marko elovainio (marko.elovainio@helsinki.fi)

Yan Huang (260401212@qq.com)

Version: 1 Date: 18 Sep 2018

Author’s response to reviews:

Dear Editors and Reviewers:

Thank you very much for your letter and for the editors’ and reviewers’ comments on our manuscript entitled “The Chinese version of Attitudes towards guidelines scale: Validity and reliability assessment” (ID: BMRM-D-18-00319). Those comments are all valuable and very helpful for revising and improving our paper, as well as the important guiding significance to our researches. We have studied comments carefully and made extensive modifications to our manuscript and supplemented extra data to make our results convincing. Revised portion are marked in red in the paper. We hope that the revision is acceptable and look forward to hearing from you soon.

With best wishes,